\documentclass[11pt]{article}
 \setlength{\textheight}{10.4in}
 \setlength{\topmargin}{-1in}
 \setlength{\headheight}{0in}
 \setlength{\topsep}{0in}
 \setlength{\marginparwidth}{0in}
 \setlength{\marginparsep}{0in}
% \setlength{\footskip}{4pc}
%\setlength{\footheight}{0in}
 \setlength{\oddsidemargin}{0in}
 \setlength{\evensidemargin}{0in}
 \setlength{\textwidth}{6.4in}
 \setlength{\parskip}{4pt}
 \setlength{\parindent}{0pt}
\parindent=0in
\parskip=1pc
\begin{document}
\thispagestyle{empty}
\section*{Quizzes}
A Quiz is a take-home problem set, one per week. They are
graded like exams, score 0-100 on each problem. The quizzes make up 10
percent of the final grade.

Quizzes are assigned weekly, and due the next week. If it is week 5 and
this message appears instead of Quiz5, then there is a mistake.
Sometimes, the missing Quiz is due to browser caching. Please reload the
PDF for the missing quiz, to insure there really is a mistake. Ask a
friend with a different computer to confirm the mistake. Then
please call, write email or ask questions in person.

\section*{Midterm Exams} They will parallel historical 2250 exams, 2014
and earlier. Sample exams will be posted when first available. There is
a rich history of midterm exams available on the course web site.


\end{document}
